Author's response to reviews

Title: Hypoglycemic effect of polysaccharides with different molecular weight of Pseudostellaria heterophylla

Authors:

Juan Hu (hujfjtcm.edu.cn)
Wensheng Pang (pws@fjtcm.edu.cn)
Jinlong chen (463618356@qq.com)
Shaowei Bai (287997445@qq.com)
Zhenzhu Zheng (smilepearl@126.com)
Xiaohua Wu (hzf_ketizu@163.com)

Version: 4 Date: 11 June 2013

Author's response to reviews: see over
Response to reviewers

Dear editor:

Thank you for your review.

We have made some changes to our manuscript. A version will be resubmitted. Responses to reviewers are as follow:

1. The animal studies had been approved by the Fujian Institute of Traditional Chinese Medicine Animal Ethics Committee, Fuzhou, China and reference number was FJZYY-AEEI 2010012), it have been added into revision in “method section”.

2. Figure legends and tables have been removed from our uploaded figures. All figures, tables have been placed the end of main manuscript.

3. Authors’ Contributions section has been corrected according to the formatting guidelines of “BMC Complementary and Alternative Medicine”.

Finally, our revised manuscript conforms to the journal style.